Author’s response to reviews

Title: Adoption and diffusion of zoning bylaws banning fast food drive-through services across Canadian municipalities

Authors:

Candace Nykiforuk (candace.nykiforuk@ualberta.ca)
Elizabeth Campbell (ejcampbe@ualberta.ca)
Soultana Macridis (soultana@ualberta.ca)
Daniel McKennitt (dwm4@ualberta.ca)
Kayla Atkey (atkey@ualberta.ca)
Kim Raine (kim.raine@ualberta.ca)

Version: 2 Date: 11 Oct 2017

Author’s response to reviews:

Dear Editorial Office,

Thank you for the notification that our map (Figure 2) required revision. We have re-created the map ourselves using ESRI ArcView software and have uploaded it as a revised Figure 2. It should now be compliant with the Creative Commons 4.0 license. Please do not hesitate to contact me if you require anything further.

Best,

Candace Nykiforuk

corresponding author